Author's response to reviews

Title: Anti-centromere antibody-seropositive Sjogren’s syndrome differs from conventional subgroup in clinical and pathological study

Authors:

Hideki Nakamura (nakamura_hideki911@yahoo.co.jp)
Atsushi Kawakami (atsushik@net.nagasaki-u.ac.jp)
Tomayoshi Hayashi (toma@net.nagasaki-u.ac.jp)
Naoki Iwamoto (naoki_iwa@hotmail.com)
Akitomo Okada (akitomoo@hotmail.com)
Mami Tamai (mmtamami@yahoo.co.jp)
Satoshi Yamasaki (syama@nagasaki-u.ac.jp)
Hiroaki Ida (idah@net.nagasaki-u.ac.jp)
Katsumi Eguchi (eguchi@net.nagasaki-u.ac.jp)

Version: 4 Date: 28 June 2010

Author's response to reviews:

According to suggestions from your editorial team, I have made next corrections.
1. Change of affiliation in full in the first page.
2. Deletion of running head, line numbers and key words
3. Add authors' name up to 30 in references.

These correction has been made in blue typing.